Author's response to reviews

Title: EBV Myelitis and Castleman's Disease in a Patient with Acquired Immune Deficiency Syndrome (AIDS): a Case Report

Authors:

Costantine Albany (calbany@iupui.edu)
George Psevdos (GPsevdos@chpnet.org)
Jasminka Balderacchi (JBaldera@chpnet.org)
Virginia Sharp (Vsharp@chpnet.org)

Version: 2 Date: 8 September 2010

Author's response to reviews: see over
Dear Editor-in-chief,

We did the following changes to our manuscript:

1- We changed the title from EBV Myelitis and Castleman’s Disease in a Patient with AIDS to EBV Myelitis and Castleman’s Disease in a Patient with Acquired Immune Deficiency Syndrome (AIDS): a Case Report

2- We change the format to fit the JMCR style.

3- We included a Consent section.

4- We added Figure legends and decreased the figure number to three.

5- We include an Authors’ Contributions section.

6- We included e-mail address for all the authors in the title page.

Thank you for considering our Manuscript for review at your journal.

If you have any question, please do not hesitate to contact me any time.

Costantine Albany